Author’s response to reviews

Title: An intervention to promote physical activity and self-management in people with stable chronic heart failure The Home-Heart-Walk study: study protocol for a randomized controlled trial

Authors:

Hui Y Du (h.du@curtin.edu.au)
Phillip J Newton (p.newton@curtin.edu.au)
Robert Zecchin (robert_zecchin@wsahs.nse.gov.au)
Alan Robert Denniss (r.denniss@uws.edu.au)
Yenna Salamonson (y.salamonson@uws.edu.au)
Bronwyn Everett (bronwyn.everett@uts.edu.au)
David C Currow (david.currow@health.sa.gov.au)
Peter S Macdonald (p.macdonald@stvincents.com.au)
Patricia M Davidson (p.davidson@curtin.edu.au)

Version: 9 Date: 21 February 2011

Author’s response to reviews: see over
Dear Editors-in-Chief:

Thank you for your time and consideration of the manuscript “An intervention to promote physical activity and self-management in people with stable chronic heart failure: The Home-Heart-Walk study: study protocol for a randomized controlled trial.”

The final version of the manuscript has been reviewed and changes are made in Track Change for your review.

I would like to sincerely thank you again for your time and patience in preparation of this manuscript. I am looking forward to hear from you.

Kind regards

HuiYun Du